Author's response to reviews

Title: Health workforce issues and the Global Fund to fight AIDS, tuberculosis and malaria: an analytical review

Authors:

Sigrid Drager (draegers@who.int)
Gulin Gedik (gedikg@who.int)
Mario R Dal Poz (dalpozm@who.int)

Version: 4 Date: 16 August 2006

Author's response to reviews:

Dear BioMed Central Editorial Production Team,

The updated manuscript version that is hereby submitted has several minor revisions: Reference numbers of the tables were adjusted and three typos corrected. As requested, the section heading "Introduction" was changed to "Background". As already explained in the previous exchange of emails, the capitalisation in the title of the article was not removed as it is derived from the name of the entity "Global Fund to Fight AIDS, Tuberculosis and Malaria".

Many thanks and kind regards

Sigrid Drager